Author's response to reviews

Title: Which family physician should I choose? The analytic hierarchy process approach in the selection of a family physician

Authors:

Emel Kuruoglu (emel.kuruoglu@deu.edu.tr)
Dilek Guldal (dilek.guldal@deu.edu.tr)
Vildan Mevsim (vildan.mevsim@deu.edu.tr)
Tolga Gunvar (tolga.gunvar@deu.edu.tr)

Version: 8
Date: 3 June 2015

Author's response to reviews: see over
Dear Editor,
Thank you for your interest in our manuscript. Revisions suggested by the reviewers are as follows:

The parts we revised in the manuscript were shown in red font above and in manuscript.

Editors comment
Please include details of any funding received in an Acknowledgment section. If there was no funding, please state this also in an Acknowledgment section.

Answer: We added knowledge.

Thank you for the revisions advice.
Sincerely yours.

Vildan MEVSİM

Corresponding Author:
Vildan Mevsim, Prof, MD, PhD
Institution: Dokuz Eylul University Medical Faculty Department of Family Medicine
Address: Dokuz Eylül Universitesi Tıp Fakültesi Aile Hekimliği AD. 35340 Balçova İzmir TURKEY
Tel: +90 532 2360044
Fax: +90 4124959
E-mail: vildan.mevsim@deu.edu.tr